ORIGINAL PAPER



Truncated Poisson–Dirichlet approximation for Dirichlet process hierarchical models

Junyi Zhang^{1,2} · Angelos Dassios¹

Received: 19 June 2022 / Accepted: 30 December 2022 $\ensuremath{\textcircled{}}$ The Author(s) 2023

Abstract

The Dirichlet process was introduced by Ferguson in 1973 to use with Bayesian nonparametric inference problems. A lot of work has been done based on the Dirichlet process, making it the most fundamental prior in Bayesian nonparametric statistics. Since the construction of Dirichlet process involves an infinite number of random variables, simulation-based methods are hard to implement, and various finite approximations for the Dirichlet process have been proposed to solve this problem. In this paper, we construct a new random probability measure called the truncated Poisson–Dirichlet process. It sorts the components of a Dirichlet process in descending order according to their random weights, then makes a truncation to obtain a finite approximation for the distribution of the Dirichlet process. Since the approximation is based on a decreasing sequence of random weights, it has a lower truncation error comparing to the existing methods using stick-breaking process. Then we develop a blocked Gibbs sampler based on Hamiltonian Monte Carlo method to explore the posterior of the truncated Poisson–Dirichlet process. This method is illustrated by the normal mean mixture model and Caron–Fox network model. Numerical implementations are provided to demonstrate the effectiveness and performance of our algorithm.

Keywords Dirichlet process \cdot Poisson–Dirichlet process \cdot Bayesian nonparametric hierarchical models \cdot Normal mean mixture models \cdot Gibbs sampling \cdot Hamiltonian Monte Carlo

1 Introduction

The Dirichlet process is a class of random probability measures; it was introduced by Ferguson (1973) to solve nonparametric inference problems from a Bayesian point of view. Since then, much research has been done based on the Dirichlet process, making its distribution a most fundamental prior in Bayesian nonparametric statistics. In the original work of Ferguson (1973), the Dirichlet process was defined by requiring its finite-dimensional marginal distribution to be a Dirichlet distribution. An equivalent representation based on the normalised jumps of a gamma process can also be

Junyi Zhang and Angelos Dassios have contributed equally to this work.

 Junyi Zhang JunyZhang@polyu.edu.hk
 Angelos Dassios
 A.Dassios@lse.ac.uk

¹ Department of Statistics, London School of Economics, Houghton Street, London WC2A 2AE, UK

² Department of Applied Mathematics, The Hong Kong Polytechnic University, Hung Hom, Hong Kong found in the same work. Later on, Sethuraman (1994) gave a constructive definition for Dirichlet process using the stickbreaking method. A more general class of random probability measures called the stick-breaking priors was studied by Ishwaran and James (2001), and the blocked Gibbs sampler for the posterior of the corresponding Bayesian hierarchical models was also developed. We refer to Yamato (2020) for a summary of different construction methods and properties of the Dirichlet process.

Due to the infinite-dimensional support of the Dirichlet process, both the stick-breaking representation and the gamma process representation involve an infinite number of random variables. This is desirable in Bayesian nonparametric statistics as it makes the prior rich enough to incorporate any given information. However, with an infinite number of random variables, it is difficult to apply simulation-based methods for posterior inference. In the existing literature, there are two approaches to work with infinite-dimensional priors, namely marginalisation and truncation.

The marginalisation approach was introduced by Escobar (1994), it marginalises the distribution of the Dirichlet process via the Blackwell–MacQueen Pólya urn scheme (Blackwell and MacQueen 1973) and leads to a powerful Markov chain Monte Carlo method. This approach is widely used in Dirichlet process hierarchical models. However, it suffers from the side effect of, taking the nonparametric hierarchical model as an example, making inference based only on the latent variables. See Ishwaran and James (2001) for a further discussion about its limitations.

The truncation approach replaces the infinite-dimensional prior with a finite-dimensional approximation. An initial study of a finite approximation of the Dirichlet process is Muliere and Tardella (1998). Using a random stopping rule, they designed an approximation method for Dirichlet process that is able to control the truncation error a priori. Ishwaran and Zarepour (2000) proposed the weak limit representation and almost sure truncation for Dirichlet process, and developed a blocked Gibbs sampler for the posterior of the corresponding hierarchical models. Ishwaran and James (2001) studied the finite approximation of the stick-breaking priors. These methods have been built based on the stickbreaking representation of Dirichlet process. On the other hand, Zarepour and Al Labadi (2012) and Argiento et al. (2016) used the gamma process representation to develop the approximation; they also analysed the truncation error of this method. A comprehensive investigation and characterisation of different truncation methods can be found in Campbell et al. (2019).

The goal of this work is to propose a new class of random probability measures called the truncated Poisson-Dirichlet process, and use it to approximate the distribution of the Dirichlet process. The name that we have chosen for the random probability measure is inspired by the work of Kingman (1975), who showed that the ranked random weights of a Dirichlet process follow the Poisson-Dirichlet distribution. Our random probability measure takes the advantage of the Poisson-Dirichlet distribution in the way that it is based on a decreasing sequence of random weights, thus leading to a lower truncation error comparing to other methods. Moreover, we group together the smaller random weights into a single term and leave it in the truncated Poisson-Dirichlet process. This differs from the existing literature where the smaller random weights are usually removed. The motivation of this construction arises from a variation of the Caron-Fox network model (Caron and Fox 2017, see also Williamson 2016) where the focus is given to the most popular nodes. We will briefly explain the application of the truncated Poisson-Dirichlet process in such models.

Then we develop a posterior sampling algorithm for the truncated Poisson–Dirichlet process. To demonstrate the usage of the algorithm, we fit the truncated Poisson–Dirichlet process into the normal mean mixture model and Caron–Fox network model, then design the posterior inference method for these models. Numerical illustrations are provided based on simulated and real-world data sets.

The paper is organised as follows. Section 2 reviews the basic properties of the Dirichlet process. Section 3 summarises some of the common approximations for the Dirichlet process. Section 4 constructs the truncated Poisson– Dirichlet process and provides a simulation algorithm for such a process. We also analyse the truncation error in this section. Section 5 develops the posterior sampling algorithm. Section 6 illustrates the application of our method in the normal mean mixture model and Caron–Fox network model. Section 7 provides the numerical illustrations. Section 8 sets out some topics for future study.

2 Construction of the Dirichlet process

In this section, we review the construction and basic properties of the Dirichlet process. Let $\alpha > 0$ be the concentration parameter and *H* be the reference distribution, a probability measure $\mathcal{P} = DP(\alpha H)$ is called a Dirichlet process if for each measurable partition (B_1, \ldots, B_k) of the sample space, the distribution

$$(\mathcal{P}(B_1),\ldots,\mathcal{P}(B_k)) \sim \operatorname{Dir}(\alpha H(B_1),\ldots,\alpha H(B_k)),$$

where Dir(...) denotes the Dirichlet distribution.

In Bayesian nonparametric statistics, it is more common to use different representations of the Dirichlet process. There are two well-known representations in literature, namely the stick-breaking representation and the gamma process representation.

The stick-breaking representation was introduced by Sethuraman (1994) as a constructive definition of the Dirichlet process, they showed that a Dirichlet process can be represented as

$$\tilde{\mathcal{P}}_{\infty}(.) = \sum_{i=1}^{\infty} \tilde{p}_i \delta_{K_i}(.), \tag{1}$$

where $\delta_{K_i}(.)$ denotes a point mass at K_i , and K_i are independent and identically distributed random variables with distribution *H*. The variables $\{\tilde{p}_i\}_{i=1}^{\infty}$ are random weights independent of K_i , such that $0 < \tilde{p}_i < 1$ and $\sum_{i=1}^{\infty} \tilde{p}_i = 1$ almost surely. They are constructed by the stick-breaking method:

$$\tilde{p}_1 = V_1$$
 and $\tilde{p}_i = (1 - V_1) \dots (1 - V_{i-1}) V_i$, (2)

for i = 2, 3, ..., where V_i are independent Beta $(1, \alpha)$ random variables for $\alpha > 0$. Throughout this paper, we will use the tilde notation to emphasise that the sequence $\{\tilde{p}_i\}_{i=1}^{\infty}$ is presented in its original order. The subscript of infinity in $\tilde{\mathcal{P}}_{\infty}$ means the expression involves infinite terms, it will be replaced by a finite number if we apply a truncation. Notice that the sequence of random weights $(\tilde{p}_1, \tilde{p}_2, ...)$ follows a GEM distribution with parameter α .

The Dirichlet process can also be represented by the jumps of a gamma process. Consider a gamma process τ_{α} with Lévy measure $v(dw) = w^{-1}e^{-w}\mathbb{1}_{(0,\infty)}(w)dw$ at time α , then τ_{α} has the Lévy–Khintchine representation

$$\mathbb{E}(\exp(-\beta\tau_{\alpha})) = \exp\left(-\alpha \int_{0}^{\infty} (1 - e^{-\beta w}) w^{-1} e^{-w} dw\right).$$

Let $\{J_i\}_{i=1}^{\infty}$ be the ranked jumps of τ_{α} such that $J_1 > J_2 > \dots$ and $\tau_{\alpha} = \sum_{i=1}^{\infty} J_i$, and let the sequence $\{p_i\}_{i=1}^{\infty}$ be defined as $p_i := J_i/\tau_{\alpha}$ for $i = 1, 2, \dots$, it follows immediately that $p_i > 0$ and $\sum_{i=1}^{\infty} p_i = 1$ with probability 1. Ferguson (1973) showed that the random probability measure

$$\mathcal{P}_{\infty}(.) = \sum_{i=1}^{\infty} p_i \delta_{K_i}(.) = \sum_{i=1}^{\infty} \frac{J_i}{\sum_{j=1}^{\infty} J_j} \delta_{K_i}(.), \qquad (3)$$

is a Dirichlet process with concentration parameter α and reference distribution *H*. Since $J_1 > J_2 > \ldots$, the random weights $\{p_i\}_{i=1}^{\infty}$ are decreasing, this differs from stick-breaking representation (1). Moreover, the sequence of random weights (p_1, p_2, \ldots) follows a Poisson–Dirichlet distribution with parameter α .

The gamma process representation (3) has an equivalent expression which is known as the Ferguson–Klass representation (Ferguson and Klass 1972):

$$\mathcal{P}_{\infty}(.) = \sum_{i=1}^{\infty} \frac{\mathrm{N}^{-1}(\Gamma_i)}{\sum_{j=1}^{\infty} \mathrm{N}^{-1}(\Gamma_j)} \delta_{K_i}(.),$$

where $\Gamma_i = E_1 + \cdots + E_i$, $E_i \sim \text{Exp}(1)$ are i.i.d. exponential random variables, and $N^{-1}(x)$ is the inverse of the tail distribution of $\alpha v(dw)$, i.e.,

$$N(x) = \alpha \int_x^\infty w^{-1} e^{-w} dw, \text{ for } x > 0.$$

In spite of the elegant expression, the inverse function $N^{-1}(x)$ is analytically intractable and an approximate evaluation is needed, see Wolpert and Ickstadt (1998) for more details.

Since the random variables K_i are independent and identically distributed, the difference between the representations (1) and (3) hinges on the orderings of their random weights. McCloskey (1965) (see also Perman et al. 1992; Pitman 1996a) showed that the random weights of the stick-breaking representation are a size-biased permutation of that of the gamma process representation, meaning that $\{\tilde{p}_i\}_{i=1}^{\infty}$ is the same sequence as $\{p_i\}_{i=1}^{\infty}$ but presented in a random order

 $(p_{\sigma_1}, p_{\sigma_2}, ...)$, where $\mathbb{P}(\sigma_1 = i) = p_i$, and for k distinct $i_1, ..., i_k$,

$$\mathbb{P}(\sigma_k = i_k \mid \sigma_1 = i_1, \dots, \sigma_{k-1} = i_{k-1}) \\ = \frac{p_{i_k}}{1 - (p_{i_1} + \dots + p_{i_{k-1}})}.$$

An index *i* with a bigger 'size' p_i tends to appear earlier in the permutation, hence the name 'size-biased permutation'. On the other hand, we revert to $\{p_i\}_{i=1}^{\infty}$ by sorting $\{\tilde{p}_i\}_{i=1}^{\infty}$ in descending order. Note that these are exactly the connections between the GEM and Poisson–Dirichlet distributions.

Although both $\bar{\mathcal{P}}_{\infty}$ and \mathcal{P}_{∞} are Dirichlet processes, they will lead to different approximation methods. In the next section, we review some of the common approximations for Dirichlet process in literature.

3 Approximation for the Dirichlet process

As we saw in the previous section, the construction of the Dirichlet process involves an infinite number of random variables, and this may cause some difficulties in the simulation of Dirichlet process based models. To avoid this problem, various random probability measures have been proposed whose distributions approximate the Dirichlet process. These random probability measures can be understood as truncated versions of the Dirichlet process. In this section, we review some of the approximation methods in literature.

The most straightforward approximation is to truncate the stick-breaking representation (1) at a positive integer N. To guarantee that the truncation leads to a well-defined probability measure, we set $V_N = 1$ in (2), then we have obtained the almost sure truncation of Dirichlet process:

$$\tilde{\mathcal{P}}_N(.) = \sum_{i=1}^N \tilde{p}_i \delta_{K_i}(.).$$
(4)

Using $\tilde{\mathcal{P}}_N$ as the prior, Ishwaran and Zarepour (2000) constructed a Bayesian semiparametric model and developed a blocked Gibbs sampler for posterior inference. The sampling algorithm is based on the results of Connor and Mosimann (1969), who showed that the posterior of $\tilde{\mathcal{P}}_N$ is a generalised Dirichlet distribution, it follows that the prior is conjugate for multinomial sampling. Therefore, we are able to sample from the posterior efficiently, and this is an advantage of the almost sure truncation. A generalisation of this model based on the stick-breaking prior can be found in Ishwaran and James (2001).

It remains to choose an adequate truncation level N. A larger N makes the approximation more accurate, but it also brings more terms into the prior, and eventually into the pos-

terior sampling algorithm. To this end, Muliere and Tardella (1998) proposed a random stopping rule for $\tilde{\mathcal{P}}_N$ which controls the approximation error a priori, Ishwaran and Zarepour (2000) derived an explicit result for the approximation error, and Griffin (2016) introduced an adaptive truncation method which allows the truncation level to be decided by the algorithm and so can avoid large errors in approximating the posterior.

Alternatively, the approximation can be obtained from the gamma process representation (3). Since the infinite sum appears in both the numerator and denominator of (3), a truncation should also be applied to both of them, such that the resulting random probability measure involves only a finite number of random variables. Zarepour and Al Labadi (2012) considered the following approximation for \mathcal{P}_{∞} :

$$\mathcal{P}_N^{\text{ZA}}(.) = \sum_{i=1}^N \frac{J_i}{\sum_{j=1}^N J_j} \delta_{K_i}(.),$$

which truncates both the numerator and denominator of (3) at the first *N* terms. The construction guarantees that \mathcal{P}_N^{ZA} is a valid probability measure. Moreover, it only involves the largest *N* jumps of the underlying gamma process; hence it is possible to sample from this random probability measure and use it in simulation. Zarepour and Al Labadi (2012) used a gamma distribution Ga(α/N , 1) to approximate the Lévy measure of the gamma process, then sampled from the jumps via the inverse transform method. Al Labadi and Zarepour (2014) generalised this idea to the two-parameter Poisson– Dirichlet process and normalised inverse-Gaussian process.

The truncation can also be made to the Lévy measure v(dw) of the underlying gamma process. Recall that a gamma process has infinite activity because its Lévy measure integrates to infinity, i.e., $\int_0^\infty w^{-1} e^{-w} dw = \infty$. By truncating the Lévy measure from below, we obtain an underlying process with finite activity. Argiento et al. (2016) considered the truncated Lévy measure

$$\rho_{\epsilon}(dw) = w^{-1} e^{-w} \mathbb{1}_{(\epsilon,\infty)}(w) dw, \text{ for } \epsilon > 0,$$

and constructed the random probability measure:

$$\tilde{\mathcal{P}}_{\epsilon}^{\text{ABG}}(.) = \sum_{i=0}^{N_{\epsilon}} \frac{\tilde{J}_{i}}{\sum_{j=0}^{N_{\epsilon}} \tilde{J}_{j}} \delta_{K_{i}}(.),$$

where $\{\tilde{J}_i\}_{i=0}^{N_{\epsilon}}$ are i.i.d. jumps from the density $\rho_{\epsilon}(dw)/(\int_{\epsilon}^{\infty} \rho_{\epsilon}(dw))$, and $N_{\epsilon} \sim \text{Pois}(\int_{\epsilon}^{\infty} \rho_{\epsilon}(dw))$ denotes the total number of jumps, which is almost surely finite. As $\epsilon \to 0$, $\tilde{\mathcal{P}}_{\epsilon}^{\text{ABG}}$ converges to the Dirichlet process in distribution; hence it provides a finite approximation for the Dirichlet process. The aim of Argiento et al. (2016) was to approximate the normalised generalised gamma (NGG) process, and the

approximation for Dirichlet process can be viewed as a special case of their results.

In Bayesian nonparametric statistics, it is possible to use the Dirichlet process prior in its original form but apply a truncation to the posterior instead. Ferguson (1973) showed that the posterior of a Dirichlet process is again a Dirichlet process with an extra point mass for each observation, this property implies a sampling algorithm for the posterior distribution which is known as the 'Chinese restaurant process' (see Pitman 2006). Since the posterior of a Dirichlet process still involves infinite terms, Gelfand and Kottas (2002) suggested to apply the truncation directly to the posterior. A generalisation of this approach for the Pitman-Yor process prior can be found in Ishwaran and James (2001), whose posterior was given by Pitman (1996b). More recently, James et al. (2009) derived a general result about the posterior of the normalised random measure with independent increment (NRMI), and Barrios et al. (2013) studied the truncation method for the posterior of the NRMIs. This approach has the advantage of using the prior in its original form, while the truncation method for posterior is similar to what we have introduced above. For example, the random stopping rule $N_{\epsilon} := \inf\{i \in \mathbb{N} \mid J_{i+1} / \sum_{j=1}^{i} J_j < \epsilon\}$ can be used to truncate the posterior of a Dirichlet process. This machinery stops sampling when the relative size of the newly sampled jump is small enough, thus involving a finite number of random variables. Argiento et al. (2010) provided an upper bound for the ignored jump sizes in this method.

As we have seen above, a lot of effort has been devoted to dealing with the problem of infinite random variables in the Dirichlet process. In this paper, we provide a new solution to this problem. Our method is inspired by the following observation. Let $\{J_i\}_{i=1}^{\infty}$ be the jumps of a gamma process τ_{α} sorted in descending order. Conditioning on the largest N jumps J_1, \ldots, J_N , the sum of the smaller jumps $\tau_{\alpha}^{N-} := \sum_{i=N+1}^{\infty} J_i$ is a truncated gamma process, and it admits the Lévy–Khintchine representation

$$\mathbb{E}\left(\exp(-\beta\tau_{\alpha}^{N-}) \mid J_{1},\ldots,J_{N}\right)$$
$$=\exp\left(-\alpha\int_{0}^{J_{N}}(1-e^{-\beta w})w^{-1}e^{-w}dw\right)$$

We view the right hand side of the equation as the Laplace transform of τ_{α}^{N-} , then derive the density of τ_{α}^{N-} by inverting this Laplace transform. In the meanwhile, the exact simulation algorithm for a truncated gamma process has been well-studied in literature. Since a truncated gamma process, we can sample from a Dickman process and use rejection sampling to include the exponential tilt. For the exact simulation algorithm of the Dickman process, see Devroye and Fawzi (2010), Fill and Huber (2010), Chi (2012) and Cloud and

Huber (2017). More recently, Dassios et al. (2019) designed an exact simulation algorithm that samples from the truncated gamma process directly. Using these results, we provide a new approximation for the Dirichlet process. We will split the random weights of \mathcal{P}_{∞} (see Eq. 3) into two parts:

i. the largest N random weights,

$$p_1 := \frac{J_1}{\sum_{j=1}^{\infty} J_j}, \quad \dots, \quad p_N := \frac{J_N}{\sum_{j=1}^{\infty} J_j};$$
 (5)

ii. the sum of the smaller random weights,

$$r := \sum_{i=N+1}^{\infty} p_i = \frac{\sum_{i=N+1}^{\infty} J_i}{\sum_{j=1}^{\infty} J_j}.$$
(6)

This is a straightforward truncation for the gamma process representation (3), without changing the total mass or Lévy measure. Moreover, we reserve the term $(\sum_{j=N+1}^{\infty} J_j)/(\sum_{J=1}^{\infty} J_j)$ of the random probability measure and explain it as an indicator for the proportion of the sample space not belonging to the largest *N* random weights. Therefore, our method can be explained as a rearrangement of the random weights according to their importance.

If we need a motivation, apart from the approximation for Dirichlet process, for our truncation method, the recent research Caron and Fox (2017) and Williamson (2016) about network model should be mentioned. Next, we briefly introduce their models with some simplifications. Let $\tilde{\mathcal{P}}_{\infty}$ be a Dirichlet process, and let *G* be a sample of $\tilde{\mathcal{P}}_{\infty}$, then *G* has the format $\sum_{i=1}^{\infty} \tilde{p}_i \delta_{K_i}(.)$. We denote by K_i the index, or identifier, of a node in the network, and \tilde{p}_i the relative popularity of this node. We also use the notation (s, r) for the links in the network, where *s* is the index of the sending node, and *r* is the index of the receiving node. Assume that each link chooses its origin and destination according to *G* independently, then a network model containing *n* links can be summarised as

$$G \sim \tilde{\mathcal{P}}_{\infty}$$

$$s_j, r_j \stackrel{\text{i.i.d.}}{\sim} G, \quad j = 1, \dots, n.$$
(7)

In other words, each link in the network is a sample of the product measure $G \times G$, and the Dirichlet process $\tilde{\mathcal{P}}_{\infty}$ can be viewed as a prior for the relative popularities of the nodes. A node with a higher relative popularity is more likely to send or receive a link, hence more active in the network. Williamson (2016) derived the explicit predictive distribution for this model using a two-dimensional version of the Pólya urn scheme.

A more general network model was considered by Caron and Fox (2017), where the popularity of the nodes are mod-

elled by the jump sizes of a gamma process. Let $\{\tilde{J}_i\}_{i=1}^{\infty}$ be the jumps of a gamma process τ_{α} without ordering, and let each node K_i have the popularity \tilde{J}_i . Assume that the links choose their origins and destinations according to $\sum_{i=1}^{\infty} (\tilde{J}_i/\tau_{\alpha}) \delta_{K_i}(.)$, which is clearly a sample of the Dirichlet process. Moreover, assume that the total number of links follows a Poisson distribution with mean τ_{α}^2 . Then the Caron–Fox network model can be written as

$$G \sim \tilde{\mathcal{P}}_{\infty}$$

$$M \sim \text{Poisson}(\tau_{\alpha}^{2})$$

$$s_{j}, r_{j} \stackrel{\text{i.i.d.}}{\sim} G = \sum_{i=1}^{\infty} \frac{\tilde{J}_{i}}{\tau_{\alpha}} \delta_{K_{i}}(.), \quad j = 1, \dots, M.$$
(8)

Conditioning on the total number of links M, model (8) reverts to model (7). However, the Poisson assumption is useful in the analysis of some real-world networks where each link has an associated count.

Next, Caron and Fox (2017) suggested to split the nodes into the following two clusters: (i) the nodes with at least one link, (ii) the nodes without any link. Then they proposed a method that estimates the popularity parameter \tilde{J}_i of each node in the former cluster, and the total popularity parameter of the nodes in the latter cluster. We emphasise that this method estimates the actual popularity parameters \tilde{J}_i , not only their relative weights $\tilde{J}_i/\tau_{\alpha}$. Hence, the model prior must be built based on the jumps of a gamma process, rather than the stick-breaking random weights \tilde{p}_i .

Inspired by the novel works of Caron and Fox (2017) and Williamson (2016), we want to design a prior that deals with the 'network with key nodes' scenario, meaning that we want to focus on the nodes with the highest popularity parameters. Think about a network where a small proportion of nodes occupy the majority of links, while most of the nodes have very few links. An example of this scenario is the social network, where a few top users enjoy the majority of followers, while most of the users have very few followers. It is natural for us to focus on the most popular users. On the other hand, there are many practical applications where only the most popular nodes are relevant. For example, in the 'who to follow suggestions' function on a social network, we only expect the most popular users to appear, and an estimation for all the users would be unnecessary and too costly. As a result, we will not estimate the popularity parameters of all the nodes, but only the most popular ones. For this purpose, we sort all the popularity parameters $\{\tilde{J}_i\}_{i=1}^{\infty}$ in descending order and focus on the largest N terms, namely J_1, \ldots, J_N . The smaller popularity parameters are grouped together in terms of $\sum_{i=N+1}^{\infty} J_i$. Given the observations from the network, we would like to estimate $(J_1, \ldots, J_N, \sum_{i=N+1}^{\infty} J_i)$. This can be done if we replace the prior of model (8) by a random probability measure with random weights in terms of (5)

and (6), and that is exactly the truncated Poisson–Dirichlet process.

4 The truncated Poisson–Dirichlet process

From the previous section, we know that it is a common trick to avoid the infinite number of random variables in Dirichlet process by ignoring the smaller random weights. In this section, we construct a new random probability measure called the truncated Poisson–Dirichlet process. This random probability measure provides a finite approximation for the distribution of Dirichlet process using the random weights in terms of (5) and (6).

Definition 1 (*Truncated Poisson–Dirichlet process*) For a positive integer N, a truncated Poisson–Dirichlet process is defined as

$$\mathcal{P}_N(.) = \sum_{i=1}^N p_i \delta_{K_i}(.) + r \delta_{K_{N+1}}(.),$$

with

$$p_i := \frac{J_i}{\sum_{j=1}^{\infty} J_j}, \text{ for } i = 1, \dots, N,$$

and

$$r := \frac{\sum_{i=N+1}^{\infty} J_i}{\sum_{j=1}^{\infty} J_j},$$

where $\{J_i\}_{i=1}^{\infty}$ denotes the ranked jumps of a gamma process τ_{α} , such that $J_1 > J_2 > \cdots$ and $\tau_{\alpha} = \sum_{i=1}^{\infty} J_i \sim \text{Ga}(\alpha, 1)$, $\alpha > 0; K_1, \ldots, K_N, K_{N+1}$ are i.i.d. random variables with distribution H, and $\delta_{K_i}(.)$ is a point mass at K_i .

When $N \to \infty$, \mathcal{P}_N converges to \mathcal{P}_∞ almost surely. Thus, it provides a finite approximation for the distribution of the Dirichlet process. The truncated Poisson–Dirichlet process still involves an infinite sum, but conditioning on J_1, \ldots, J_N , the sum of smaller jumps $\sum_{i=N+1}^{\infty} J_i$ is a truncated gamma process. To derive its density, we first prepare the following lemma.

Lemma 1 (Density of a truncated gamma process) Let $Z_{\alpha,\mu}$ be a truncated gamma process with the Lévy–Khintchine representation

$$\mathbb{E}(\exp(-\beta Z_{\alpha,\mu})) = \exp\left(-\alpha \int_0^1 (1 - e^{-\beta w}) w^{-1} e^{-\mu w} dw\right),$$

and let $g_Z(z; \alpha, \mu)$ be the density of $Z_{\alpha,\mu}$, then

$$g_Z(z;\alpha,\mu) = \sum_{i=0}^{n-1} \frac{(-\alpha)^i}{i!} L_i(z)$$

for $n - 1 < z \le n$, n = 1, 2, ..., where $L_i(t)$ is defined recursively as follows:

$$L_0(z) = e^{\alpha \Gamma(0,\mu)} \frac{\mu^{\alpha}}{\Gamma(\alpha)} z^{\alpha-1} e^{-\mu z}, \text{ for } z > 0,$$

and

$$L_{i+1}(z) = \int_{1}^{z-i} L_i(z-s)s^{-1}e^{-\mu s}ds, \text{ for } z > i+1$$

Proof The density of $Z_{\alpha,\mu}$ can be derived via inverse Laplace transform, we have

$$g_{Z}(z; \alpha, \mu)$$

$$= \mathcal{L}^{-1} \{ \mathbb{E}(\exp(-\beta Z_{\alpha,\mu})) \}$$

$$= \mathcal{L}^{-1} \{ \exp\left(-\alpha \int_{0}^{1} (1 - e^{-\beta w}) w^{-1} e^{-\mu w} dw\right) \}$$

$$= \mathcal{L}^{-1} \{ \exp\left(-\alpha \int_{0}^{\infty} (1 - e^{-\beta w}) w^{-1} e^{-\mu w} dw\right) \}$$

$$\times \exp\left(\alpha \int_{1}^{\infty} (1 - e^{-\beta w}) w^{-1} e^{-\mu w} dw\right) \}$$

$$= \mathcal{L}^{-1} \{ (1 + \beta/\mu)^{-\alpha} \exp\left(\alpha \int_{1}^{\infty} w^{-1} e^{-\mu w} dw\right) \}$$

$$= \mathcal{L}^{-1} \{ (1 + \beta/\mu)^{-\alpha} e^{\alpha \Gamma(0,\mu)}$$

$$\times \sum_{k=0}^{\infty} \frac{(-\alpha)^{k}}{k!} \left(\int_{1}^{\infty} e^{-\beta w} w^{-1} e^{-\mu w} dw \right)^{k} \},$$
(9)

where $\Gamma(0, \mu)$ is the exponential integral

$$\Gamma(0,\mu) = \int_{\mu}^{\infty} w^{-1} e^{-w} dw.$$

From the Laplace transform of a gamma random variable, we know

$$\mathcal{L}^{-1}\left\{\frac{e^{\alpha\Gamma(0,\mu)}}{(1+\beta/\mu)^{\alpha}}\right\} = e^{\alpha\Gamma(0,\mu)}\frac{\mu^{\alpha}}{\Gamma(\alpha)}z^{\alpha-1}e^{-\mu z},$$

denote this function by $L_0(z)$, then the last line of (9) can be inverted by convolution.

To illustrate the result of Lemma 1, we plot the density $g_Z(z; \alpha, \mu)$ with different values of α and μ in Fig. 1. We



Fig. 1 Density of the truncated gamma process and the numerical inverse Laplace transform results

also apply the CME method (see Horváth et al. 2020) to invert the Laplace transform of $Z_{\alpha,\mu}$ numerically at some fixed points, and plot the results as stars. From the figure we can see that both methods produce similar results.

Now we can use Lemma 1 to derive the conditional density of $\sum_{i=N+1}^{\infty} J_i$ given J_1, \ldots, J_N .

Corollary 2 Let $\{J_i\}_{i=1}^{\infty}$ be the ranked jumps of a gamma process τ_{α} . Denote by $\tau_{\alpha}^{N-} := \sum_{i=N+1}^{\infty} J_i$ the sum of the smaller jumps, then τ_{α}^{N-} has the conditional Lévy–Khintchine representation

$$\mathbb{E}\left(\exp(-\beta\tau_{\alpha}^{N-})\mid J_{1},\ldots,J_{N}\right)$$
$$=\exp\left(-\alpha\int_{0}^{J_{N}}(1-e^{-\beta w})w^{-1}e^{-w}dw\right).$$

It follows that the conditional density of τ_{α}^{N-} , given J_1, \ldots, J_N , is

$$f_{\alpha,J_N}(w)dw = J_N^{-1}g_Z\bigg(wJ_N^{-1};\alpha,J_N\bigg)dw,$$

where $g_Z(z; \alpha, \mu)$ is provided in Lemma 1.

Proof The Lévy–Khintchine representation of τ_{α}^{N-} follows from the basic properties of Poisson random measure, see Kyprianou (2006). It can be written as

$$\mathbb{E}\left(\exp(-\beta\tau_{\alpha}^{N-})\mid J_{1},\ldots,J_{N}\right)$$
$$=\exp\left(-\alpha\int_{0}^{1}(1-e^{-\beta J_{N}w})w^{-1}e^{-J_{N}w}dw\right),$$

then the conditional density of τ_{α}^{N-} is a direct consequence of Lemma 1.

Then we can derive the joint density of the jump sizes of the gamma process used in Definition 1.

Theorem 3 Let $\{J_i\}_{i=1}^{\infty}$ be the ranked jumps of a gamma process τ_{α} with Lévy measure $w^{-1}e^{-w}\mathbb{1}_{(0,\infty)}(w)dw$ at time α . Denote by $R_k := J_{k+1}/J_k$ the ratio between the (k + 1)-th and k-th largest jumps, and set $\tau_{\alpha}^{N-} := \sum_{i=N+1}^{\infty} J_i$, then the joint density of the random vector $(J_1, R_1, \ldots, R_{N-1}, \tau_{\alpha}^{N-})$ is

$$\mathbb{P}(J_{1} \in dx_{1}, R_{1} \in dr_{1}, \dots, R_{N-1} \in dr_{N-1}, \tau_{\alpha}^{N-} \in dy)$$

$$= \exp\left(-\alpha \int_{x_{1}r_{1}\dots r_{N-1}}^{\infty} w^{-1}e^{-w}dw\right)$$

$$\times \frac{\alpha^{N}}{x_{1}r_{1}\dots r_{N-1}}e^{-x_{1}(1+r_{1}+r_{1}r_{2}+\dots+r_{1}r_{2}\dots r_{N-1})}$$

$$\times f_{\alpha,x_{1}r_{1}\dots r_{N-1}}(y)dydr_{N-1}\dots dr_{1}dx_{1},$$

where $x_1 \in (0, \infty)$, $r_k \in (0, 1)$ for k = 1, ..., N - 1, $y \in (0, \infty)$, and $f_{\alpha, x_1 r_1 ... r_{N-1}}(y)$ is the density of a truncated gamma process with the Lévy measure $w^{-1}e^{-w} \mathbb{1}_{(0, x_1 r_1 ... r_{N-1})}(w) dw$ at time α .

Proof The joint density of the largest N jumps of a gamma process follows from the basic properties of Poisson random measure (see Kyprianou 2006):

$$\mathbb{P}(J_1 \in dx_1, \dots, J_N \in dx_N)$$

= $\exp\left(-\alpha \int_{x_N}^{\infty} w^{-1} e^{-w} dw\right)$
 $\times \frac{\alpha^N}{x_1 \dots x_N} e^{-(x_1 + \dots + x_N)} dx_N \dots dx_1,$

where $x_1 > x_2 > \cdots > x_N > 0$. Then the joint density of $(J_1, R_1, \ldots, R_{N-1})$ can be obtained by changing the variables with $x_k = x_1 \prod_{i=1}^{k-1} r_i$, for $k = 2, \ldots, N$. The conditional density of $\sum_{i=N+1}^{\infty} J_i$ is given by Corollary 2, and this completes the proof.

Next, we provide a simulation algorithm for the random vector $(J_1, \ldots, J_N, \tau_\alpha^{N-})$. The algorithm consists two stages; we first applied the rejection method introduced by Rosiński (2001) (see also Godsill and K1 ndap 2022; Imai and Kawai 2013) to sample from the largest *N* jumps (J_1, \ldots, J_N) of the gamma process, then use the exact simulation algorithm for truncated gamma process in Dassios et al. (2019) to simulate the sum of smaller jumps τ_α^{N-} . We refer to Algorithm 3.2 of Dassios et al. (2019) as AlgorithmTG(*t*, μ), and attach the full steps in Appendix A.

Proof of Algorithm 1 We apply the rejection method to sample from (J_1, \ldots, J_N) . This method uses a dominating point process X_{α} with Lévy measure $v_0(dw) := w^{-1}(1+w)^{-1}dw$ at time α . Since $v_0(dw)$ has the tail probability $\int_x^{\infty} v_0(dw) =$

Statistics and Computing (2023) 33:30

 Table 1
 Sample averages of the largest 5 jumps of a gamma process, the sample size is 10⁴ and CPU is at 2.80 GHz

| | Algorithm | J_1 | J_2 | J_3 | J_4 | J_5 | τ_{α}^{5-} | Sum | Time (s) |
|----------------|-------------|--------|--------|--------|--------|--------|----------------------|--------|----------|
| $\alpha = 0.5$ | True mean | 0.3788 | 0.0855 | 0.0245 | 0.0076 | 0.0024 | 0.0012 | 0.5000 | N.A. |
| | HMC | 0.3772 | 0.0842 | 0.0249 | 0.0086 | 0.0027 | 0.0014 | 0.4990 | 72.7881 |
| | Algorithm 1 | 0.3758 | 0.0848 | 0.0243 | 0.0077 | 0.0025 | 0.0012 | 0.4963 | 0.5651 |
| $\alpha = 1.0$ | True mean | 0.6250 | 0.2097 | 0.0884 | 0.0404 | 0.0192 | 0.0183 | 1.0000 | N.A. |
| | HMC | 0.6108 | 0.2066 | 0.0840 | 0.0380 | 0.0174 | 0.0170 | 0.9739 | 98.5934 |
| | Algorithm 1 | 0.6248 | 0.2097 | 0.0886 | 0.0405 | 0.0191 | 0.0182 | 1.0009 | 0.5950 |
| $\alpha = 1.5$ | True mean | 0.8066 | 0.3241 | 0.1624 | 0.0881 | 0.0494 | 0.0694 | 1.5000 | N.A. |
| | HMC | 0.7983 | 0.3124 | 0.1547 | 0.0833 | 0.0449 | 0.0639 | 1.4575 | 122.4333 |
| | Algorithm 1 | 0.8066 | 0.3242 | 0.1624 | 0.0882 | 0.0496 | 0.0695 | 1.5007 | 0.6864 |
| $\alpha = 2.0$ | True mean | 0.9519 | 0.4246 | 0.2361 | 0.1392 | 0.0882 | 0.1593 | 2.0000 | N.A. |
| | HMC | 0.9549 | 0.3981 | 0.2204 | 0.1419 | 0.0845 | 0.1553 | 1.9550 | 133.8877 |
| | Algorithm 1 | 0.9564 | 0.4310 | 0.2378 | 0.1423 | 0.0882 | 0.1620 | 2.0178 | 0.7195 |
| | | | | | | | | | |

Algorithm 1 Simulation algorithm for the random vector $(J_1, \ldots, J_N, \tau_{\alpha}^{N-})$

1: Set $k \leftarrow 1, i \leftarrow 1, \Gamma_0 \leftarrow 0$. 2: while $k \leq N$ do 3: Set $\Gamma_i \leftarrow \Gamma_{i-1} + \operatorname{Exp}(1)$. Set $x_i \leftarrow (\exp(\Gamma_i/\alpha) - 1)^{-1}$. 4: 5: Set $U_i \leftarrow \text{Uniform}[0, 1]$. 6: if $U_i \leq (1 + x_i) \exp(-x_i)$ then 7: Set $J_k \leftarrow x_i$ and $k \leftarrow k+1$. 8: end if 9: Set $i \leftarrow i + 1$. 10: end while 11: Set $\tau_{\alpha}^{N-} \leftarrow J_N \times \text{AlgorithmTG}(\alpha, J_N)$. 12: Output $(J_1, ..., J_N, \tau_{\alpha}^{N-})$.

 $\alpha \ln(x^{-1} + 1)$, which can be inverted explicitly, we sample from the ranked jumps of X_{α} via Step 4. The *i*th largest jump of X_{α} is accepted with probability $(1 + x_i) \exp(-x_i)$. We refer to in Rosiński (2001), Godsill and K1 ndap (2022) and Imai and Kawai (2013) for more details.

Conditioning on (J_1, \ldots, J_N) , the Lévy–Khintchine representation of τ_{α}^{N-} is (see the proof of Corollary 2):

$$\mathbb{E}\left(\exp(-\beta\tau_{\alpha}^{N-})\mid J_{1},\ldots,J_{N}\right)$$
$$=\exp\left(-\alpha\int_{0}^{1}(1-e^{-\beta J_{N}w})w^{-1}e^{-J_{N}w}dw\right),$$

it follows that $\tau_{\alpha}^{N-} \stackrel{d}{=} J_N \times Z_{\alpha, J_N}$, where $Z_{t,\mu}$ denotes a truncated gamma process with Lévy measure $w^{-1}e^{-\mu w}\mathbb{1}_{\{0 < w < 1\}}$ dw at time *t*, and the exact simulation algorithm for $Z_{t,\mu}$ can be found in Appendix A, see Dassios et al. (2019) for a further discussion.

We present the numerical results of Algorithm 1 with N = 5 and various values for α in Table 1. We also use the Hamiltonian Monte Carlo method (Neal 2011, to be introduced in Sect. 5) to sample from the joint density given in

Theorem 3. From the table we can see that Algorithm 1 simulates the jumps of the gamma process exactly, and it is very efficient. In fact, Rosiński (2001) gave that the average number of rejected jumps of the dominating process X_{α} is $\alpha\gamma$, where γ is the Euler Mascheroni constant. Therefore, to sample from the largest N jumps of the gamma process, we only need to generate $(N + \alpha\gamma)$ number of exponential and uniform random variables respectively in average. Moreover, Algorithm 1 simulates the jumps in succession, thus it is very easy to include a random stopping rule. For example, to apply the random truncation level N_{ϵ} introduced in Sect. 3, we only need to replace the fixed truncation level N by N_{ϵ} in the algorithm.

To close this section, we analyse the truncation error of the truncated Poisson–Dirichlet process. From Definition 1, we can see that the truncation error is revealed by the tail probability of the Poisson–Dirichlet process. Let $\{p_i\}_{i=1}^{\infty}$ be the random weights of a Poisson–Dirichlet process, we are interested in the tail probability defined as

$$r = \sum_{i=N+1}^{\infty} p_i = \frac{\sum_{i=N+1}^{\infty} J_i}{\sum_{j=1}^{\infty} J_j}$$
$$= \frac{\sum_{i=N+1}^{\infty} J_i}{J_1 + \dots + J_N + \sum_{j=N+1}^{\infty} J_j}.$$

Conditioning on J_1, \ldots, J_N , we can express the tail probability in terms of

$$\{r = y \mid J_1, \dots, J_N\} = \left\{ \sum_{i=N+1}^{\infty} J_i = \frac{y}{1-y} (J_1 + \dots + J_N) \mid J_1, \dots, J_N \right\}.$$

Since the densities of $J_1, ..., J_N$ and $\sum_{i=N+1}^{\infty} J_i$ have been given in Theorem 3 and Corollary 2, we can integrate out the condition to derive the explicit distribution of *r*. However, the



Fig. 2 Comparison between the tail probabilities of the Dirichlet and Poisson–Dirichlet processes

integral is too complicated to calculate, and we will provide some numerical results for the tail probability instead, see Fig. 2.

We compare the numerical results to the truncation error of the almost sure truncation, for which an explicit result can be found in Ishwaran and Zarepour (2000) (see also Ishwaran and James 2001). Let $\{\tilde{p}_i\}_{i=1}^{\infty}$ be the random weights of a Dirichlet process defined as (2), for each non-negative integer *N* and each positive integer *r*, they showed that

$$\mathbb{E}\left(\left(\sum_{i=N+1}^{\infty}\tilde{p}_i\right)^r\right) = \left(\frac{\alpha}{\alpha+r}\right)^N,\tag{10}$$

and

$$\mathbb{E}\left(\sum_{i=N+1}^{\infty}\tilde{p}_{i}^{r}\right) = \left(\frac{\alpha}{\alpha+r}\right)^{N}\frac{\Gamma(r)\Gamma(\alpha+1)}{\Gamma(\alpha+r)}.$$
(11)

These results provide an upper bound for the tail probability of the Poisson–Dirichlet process. It is clear from the definition that $\sum_{i=1}^{N} p_i \ge \sum_{i=1}^{N} \tilde{p}_i$ almost surely for any fixed N, then $\sum_{i=N+1}^{\infty} p_i \le \sum_{i=N+1}^{\infty} \tilde{p}_i$, and $(\sum_{i=N+1}^{\infty} p_i)^r \le$ $(\sum_{i=N+1}^{\infty} \tilde{p}_i)^r$. Combine this with (10), we have that

$$\mathbb{E}\left(\left(\sum_{i=N+1}^{\infty} p_i\right)^r\right) \le \left(\frac{\alpha}{\alpha+r}\right)^N.$$

Moreover, since $p_1 > p_2 > \cdots$, the sequence $\{p_i^r\}_{i=1}^{\infty}$ is decreasing, i.e., $p_1^r > p_2^r > \cdots$, it follows that $\sum_{i=1}^{N} p_i^r \ge \sum_{i=1}^{N} \tilde{p}_i^r$. On the other hand, the event $\{\sum_{i=1}^{\infty} p_i^r = \sum_{i=1}^{\infty} \tilde{p}_i^r\}$ must be true, then we deduce that $\sum_{i=N+1}^{\infty} p_i^r \le \sum_{i=N+1}^{\infty} p_i^r$

 $\sum_{i=N+1}^{\infty} \tilde{p}_i^r$. Combining this with (11), we get

$$\mathbb{E}\left(\sum_{i=N+1}^{\infty} p_i^r\right) \le \left(\frac{\alpha}{\alpha+r}\right)^N \frac{\Gamma(r)\Gamma(\alpha+1)}{\Gamma(\alpha+r)}$$

In Fig. 2, we plot the sample average of the tail probability of the Poisson–Dirichlet process, as well as the explicit tail probability of the Dirichlet process. The figure shows that we can use the truncated Poisson–Dirichlet process with values of α up to 10 and still have a reasonable truncation error if the truncation level is high enough, for example N = 50. Also, the figure implies a smaller truncation error arising from the truncated Poisson–Dirichlet process, comparing to the almost sure truncation.

5 Posterior of the truncated Poisson–Dirichlet process

In this section, we develop the posterior sampling algorithm for the truncated Poisson–Dirichlet process. We first work out the posterior, which follows from the Bayes' Theorem.

Proposition 4 For a sequence of i.i.d. observations $X = (X_1, ..., X_n)$ from a truncated Poisson–Dirichlet process P with concentration parameter α , if $n_j = card\{X_i = K_j\}$ is the number of X_i s which equal K_j , for j = 1, ..., N + 1, then the posterior is proportional to

$$\pi(P, \alpha \mid X) \propto \frac{x_1^{n_1+n_2+\dots+n_N} r_1^{n_2+\dots+n_N} \dots r_{N-1}^{n_N} y^{n_{N+1}}}{(\Sigma+y)^n} \times \mathbb{P}(J_1 \in dx_1, R_1 \in dr_1, \dots, R_{N-1} \in dr_{N-1}, \tau_{\alpha}^{N-1} \in dy) \pi(\alpha),$$
(12)

where $n = \sum_{j=1}^{N+1} n_j$ is the total number of observations, $\Sigma := x_1(1 + r_1 + \dots + r_1r_2 \dots r_{N-1})$ denotes the sum of the largest N jumps of the prior, the joint density of $(J_1, R_1, \dots, R_{N-1}, \tau_{\alpha}^{N-1})$ is given in Theorem 3, and $\pi(\alpha)$ denotes the prior of the hyperparameter α .

Proof Conditioning on the truncated Poisson–Dirichlet process prior, the likelihood function follows a multinomial distribution,

$$\mathbb{P}(\mathbf{K} \mid P, \alpha) = \left(\frac{x_1}{\Sigma + y}\right)^{n_1} \left(\frac{x_1 r_1}{\Sigma + y}\right)^{n_2} \dots \times \left(\frac{x_1 r_1 r_2 \dots r_{N-1}}{\Sigma + y}\right)^{n_N} \left(\frac{y}{\Sigma + y}\right)^{n_{N+1}};$$

then the proposition is a direct consequence of Bayes' Theorem. $\hfill \Box$

To sample from the conditional distribution (12), one way is to sample from $\pi(P \mid \mathbf{K}, \alpha)$ for a fixed α , then update α according to the conditional density $\pi(\alpha \mid P)$ with conjugate sampling or a Metropolis step. This approach was taken for the almost sure truncation of the Dirichlet process, see Ishwaran and Zarepour (2000). If we choose the same approach for the truncated Poisson–Dirichlet process, we can sample from (12) for a fixed α by rejection sampling. The joint density of $(J_1, R_1, \ldots, R_{N-1}, \tau_{\alpha}^{N-1})$ is taken to be the envelope because its exact simulation algorithm has been developed in Theorem 1. However, this method might be inefficient because the ratio between the target density and envelope is maximised at

$$\left(\frac{x_1}{\Sigma+y}\right)^{n_1} \left(\frac{x_1r_1}{\Sigma+y}\right)^{n_2} \cdots \\ \times \left(\frac{x_1r_1r_2\dots r_{N-1}}{\Sigma+y}\right)^{n_N} \left(\frac{y}{\Sigma+y}\right)^{n_{N+1}} \\ \le \left(\frac{n_1}{n}\right)^{n_1} \left(\frac{n_2}{n}\right)^{n_2} \cdots \left(\frac{n_N}{n}\right)^{n_N} \left(\frac{n_{N+1}}{n}\right)^{n_{N+1}},$$

when n is large, the acceptance rate could be low, and a lot of iterations are needed until a candidate is accepted.

For an efficient exploration of $\pi(P, \alpha \mid \mathbf{K})$, we use the Hamiltonian Monte Carlo method (see Neal 2011, see also Caron and Fox 2017 Section 7.2 for a similar application in posterior sampling) within the Gibbs sampler to update $(x_1, r_1, \ldots, r_{N-1})$. The HMC method updates the variables according to the gradient of the log-posterior, hence leading to a higher acceptance rate than the rejection sampling method described above. For the update of the remaining mass *y* and hyperparameter α , we apply a rejection sampling step. This step makes use of the exact simulation algorithm for truncated gamma process, see Appendix A. In conclusion, the sampling algorithm for (12) is given as follows.

Step 1 update $(x_1, r_1, ..., r_{N-1})$ given the rest using the HMC method;

Step 2 update the remaining mass y and hyperparameter α given the rest by rejection sampling.

To facilitate the HMC method, we express the posterior in terms of unconstrained variables. Since $x_1 \in (0, \infty)$ and $r_i \in (0, 1)$, we define

$$X := \log(x_1) \text{ and } R_i := \tan(\pi(r_i - 0.5)),$$
(13)

for i = 1, ..., N - 1, such that both X and R_i are unconstrained. Then we rewrite the posterior as

$$\pi(X, R_1, \dots, R_{n-1} | y, \alpha, \mathbf{K})$$

$$\propto (e^X)^{n_1 + n_2 + \dots + n_N}$$

$$\times \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_1)\right)^{n_2 + \dots + n_N} \dots$$

$$\times \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_{N-1})\right)^{n_N} (\Sigma + y)^{-n}$$
$$\times \exp\left(-\alpha \int_{\Pi}^{\infty} w^{-1} e^{-w} dw\right) \frac{1}{\Pi} e^{-\Sigma} f_{\alpha,\Pi}(y)$$
$$\times e^X \left(\frac{1}{\pi} \frac{1}{1+R_1^2}\right) \dots \left(\frac{1}{\pi} \frac{1}{1+R_{N-1}^2}\right)$$
$$\times dR_{N-1} \dots dR_1 dX,$$

where

$$\Pi := e^{X} \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_{1}) \right) \times \dots$$
$$\times \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_{N-1}) \right),$$

and

$$\Sigma := e^{X} \left[1 + \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_{1}) \right) + \cdots + \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_{1}) \right) \times \cdots \right] \times \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_{N-1}) \right) \right].$$

And the log-posterior is given by

$$\begin{split} \log(\pi(X, R_1, \dots, R_{n-1} \mid y, \alpha, \mathbf{K})) \\ &= C + (n_1 + n_2 + \dots + n_N)X \\ &+ (n_2 + \dots + n_N) \log\left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_1)\right) + \dots \\ &+ n_N \log\left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_{N-1})\right) \\ &- n \log(\Sigma + y) \\ &- \alpha \int_{\Pi}^{\infty} w^{-1} e^{-w} dw - \log(\Pi) - \Sigma \\ &+ \log(f_{\alpha, \Pi}(y)) + X \\ &+ \log\left(\frac{1}{\pi} \frac{1}{1 + R_1^2}\right) + \dots + \log\left(\frac{1}{\pi} \frac{1}{1 + R_{N-1}^2}\right), \end{split}$$

where C is a scaling constant independent of X, R_1, \ldots, R_{N-1} .

Next, we need to calculate the gradient of the log-posterior. The calculation is straightforward except for the derivative of $\log(f_{\alpha,\Pi}(y))$ with respect to *X* and *R_i*. To this end, we first calculate the derivative of a truncated gamma density $f_{\alpha,x}(y)$ as follows.

When $0 < y \le x$, the ratio $0 < y/x \le 1$. From Lemma 1 and Corollary 2 we know the density involves only one term,

that is,

$$f_{\alpha,x}(y) = \frac{1}{x} g_Z\left(\frac{y}{x}; \alpha, x\right) = e^{\alpha \Gamma(0,x)} \frac{1}{\Gamma(\alpha)} y^{\alpha-1} e^{-y},$$

then the derivative is

$$\frac{d}{dx}f_{\alpha,x}(y) = -\alpha x^{-1}e^{-x}e^{\alpha\Gamma(0,x)}\frac{1}{\Gamma(\alpha)}y^{\alpha-1}e^{-y}$$
$$= -\alpha x^{-1}e^{-x}f_{\alpha,x}(y).$$

Next, we consider the range $x < y \le 2x$. In this case, $1 < y/x \le 2$, and the density involves two terms,

$$f_{\alpha,x}(y) = e^{\alpha \Gamma(0,x)} \frac{1}{\Gamma(\alpha)} y^{\alpha-1} e^{-y} - \alpha \int_{1}^{y/x} e^{\alpha \Gamma(0,x)} \frac{1}{\Gamma(\alpha)} (y-xs)^{\alpha-1} e^{-y} s^{-1} ds,$$

then the derivative is

$$\begin{aligned} \frac{d}{dx}f_{\alpha,x}(y) \\ &= -\alpha x^{-1}e^{-x} \left(e^{\alpha\Gamma(0,x)} \frac{1}{\Gamma(\alpha)} y^{\alpha-1} e^{-y} \right. \\ &\left. -\alpha \int_{1}^{y/x} e^{\alpha\Gamma(0,x)} \frac{1}{\Gamma(\alpha)} (y-xs)^{\alpha-1} e^{-y} s^{-1} ds \right) \\ &\left. + \alpha e^{\alpha\Gamma(0,x)} \frac{1}{\Gamma(\alpha)} e^{-y} x^{-1} (y-x)^{\alpha-1} \right. \end{aligned}$$

The first term of this derivative can be written as $-\alpha x^{-1}e^{-x}$ $f_{\alpha,x}(y)$, while for the second term, we have

$$\alpha e^{\alpha \Gamma(0,x)} \frac{1}{\Gamma(\alpha)} e^{-y} x^{-1} (y-x)^{\alpha-1}$$

= $\alpha x^{-1} e^{-x} e^{\alpha \Gamma(0,x)} \frac{1}{\Gamma(\alpha)} (y-x)^{\alpha-1} e^{-(y-x)}$
= $\alpha x^{-1} e^{-x} f_{\alpha,x} (y-x) \mathbb{1}_{\{y>x\}}.$

Thus, the derivative is given by

$$\frac{d}{dx}f_{\alpha,x}(y) = -\alpha x^{-1}e^{-x}\left(f_{\alpha,x}(y) - f_{\alpha,x}(y-x)\mathbb{1}_{\{y>x\}}\right).$$
(14)

In general, it can be shown by induction that the derivative has the format (14) for any $(k-1)x < y \le kx, k = 1, 2, ...$ However, the calculation is quite complicated, so we provide an inverse Laplace transform method towards this result. From Lemma 1 and Corollary 2 we can see that in each subinterval $(k - 1)x < y \le kx$, the functions $f_{\alpha,x}(y)$ and $\frac{d}{dx} f_{\alpha,x}(y)$ are continuous in y and x, we then have for each subinterval that

$$\mathcal{L}^{-1} \left\{ \int_{0}^{\infty} e^{-\beta y} \frac{d}{dx} f_{\alpha,x}(y) dy \right\} \mathbb{1}_{\{(k-1)x < y \le kx\}}$$

$$= \mathcal{L}^{-1} \left\{ \frac{d}{dx} \int_{0}^{\infty} e^{-\beta y} f_{\alpha,x}(y) dy \right\} \mathbb{1}_{\{(k-1)x < y \le kx\}}$$

$$= \mathcal{L}^{-1} \left\{ -\alpha x^{-1} e^{-x} (1 - e^{-\beta x}) \times \exp\left(-\alpha \int_{0}^{x} (1 - e^{-\beta w}) w^{-1} e^{-w} dw\right) \right\}$$

$$\times \mathbb{1}_{\{(k-1)x < y \le kx\}}$$

$$= -\alpha x^{-1} e^{-x} \left(f_{\alpha,x}(y) - f_{\alpha,x}(y - x) \mathbb{1}_{\{y > x\}} \right)$$

$$\times \mathbb{1}_{\{(k-1)x < y \le kx\}}.$$

Then the derivative of $\log(f_{\alpha,x}(y))$ is

$$\frac{d}{dx}\log(f_{\alpha,x}(y))$$

= $-\alpha x^{-1}e^{-x}\left(1 - \frac{f_{\alpha,x}(y-x)\mathbb{1}_{\{y>x\}}}{f_{\alpha,x}(y)}\right)$

and this enables us to calculate the derivative of $\log(f_{\alpha,\Pi}(y))$ with respect to *X* and *R_i*,

$$\frac{d}{dX}\log(f_{\alpha,\Pi}(y)) = -\alpha e^{-\Pi} \left(1 - \frac{f_{\alpha,\Pi}(y-\Pi)\mathbb{1}_{\{y>\Pi\}}}{f_{\alpha,\Pi}(y)}\right),$$

and

$$\frac{d}{dR_i} \log(f_{\alpha,\Pi}(y))$$

$$= -\alpha e^{-\Pi} \left(1 - \frac{f_{\alpha,\Pi}(y - \Pi) \mathbb{1}_{\{y > \Pi\}}}{f_{\alpha,\Pi}(y)} \right)$$

$$\times \frac{1}{\pi (1 + R_i^2)} \left(\frac{1}{2} + \frac{1}{\pi} \arctan(R_i) \right)^{-1}$$

Now we are ready to use the HMC method to update the variables $(X, R_1, \ldots, R_{N-1})$. Let $L \ge 1$ be the number of leapfrog steps and $\epsilon > 0$ be the step size. For a concise presentation of the algorithm steps, we denote by $W_{1:N} := (X, R_1, \ldots, R_{N-1})$ the variables, and $D(W_{1:N}) :=$ $D(X, R_1, \ldots, R_{N-1}) = \nabla \log(\pi(X, R_1, \ldots, R_{n-1} | y, \alpha, \mathbf{K}))$ the gradient of the log-posterior. The HMC method samples the momentum variables $p \sim \mathcal{N}(0, I_N)$, then simulate *L* steps of the Hamiltonian via the following steps:

$$\tilde{p}^{(0)} \leftarrow p + (\epsilon/2)D(W_{1:N}), \\ \tilde{W}^{(0)}_{1:N} \leftarrow W_{1:N};$$

and for l = 1, ..., L - 1,

$$\begin{split} \tilde{W}_{1:N}^{(l)} &\leftarrow \tilde{W}_{1:N}^{(l-1)} + \epsilon \, \tilde{p}^{(l-1)}, \\ \tilde{p}^{(l)} &\leftarrow \tilde{p}^{(l-1)} + \epsilon \, D(\tilde{W}_{1:N}^{(l)}); \end{split}$$

finally, set

$$\begin{split} \tilde{W}_{1:N} &\leftarrow \tilde{W}_{1:N}^{(L-1)} + \epsilon \tilde{p}^{(L-1)}, \\ \tilde{p} &\leftarrow - \bigg\{ \tilde{p}^{(L-1)} + (\epsilon/2) D(\tilde{W}_{1:N}) \bigg\}. \end{split}$$

The proposal $\tilde{W}_{1:N} = (\tilde{X}, \tilde{R}_1, \dots, \tilde{R}_{N-1})$ is accepted if

$$U[0,1] \le \frac{\pi(\tilde{W}_{1:N} \mid y, \alpha, \mathbf{K})}{\pi(W_{1:N} \mid y, \alpha, \mathbf{K})} \exp\left(-\frac{1}{2} \sum_{i=1}^{N} (\tilde{p_i}^2 - p_i^2)\right).$$

We then output $(x_1, r_1, ..., r_{N-1})$ using the inverse transform of (13), and Step 1 is completed.

For Step 2, we need to update the remaining mass y and the hyperparameter α . We will use the acceptance–rejection method to sample from the conditional density of y and α :

$$\begin{aligned} \pi(\mathbf{y}, \boldsymbol{\alpha} \mid x_1, r_1, \dots, r_{N-1}, \mathbf{K}) \\ &= C \frac{y^{n_{N+1}}}{(\Sigma + y)^n} \exp\left(-\alpha \int_{x_1 r_1 \dots r_{N-1}}^{\infty} w^{-1} e^{-w} dw\right) \\ &\times \alpha^N f_{\alpha, x_1 r_1 \dots r_{N-1}}(y) \pi(\boldsymbol{\alpha}) \\ &= C' \frac{y^{n_{N+1}}}{(\Sigma + y)^n} \\ &\times \exp\left(-\alpha \left(v_2 + \int_{x_1 r_1 \dots r_{N-1}}^{\infty} w^{-1} e^{-w} dw\right)\right) \right) \\ &\times \alpha^{N+\nu_1-1} f_{\alpha, x_1 r_1 \dots r_{N-1}}(y), \end{aligned}$$

where $\alpha, y \in (0, \infty)$, *C* and *C'* are scaling constants, and $\pi(\alpha) \sim \text{Ga}(\nu_1, \nu_2)$ is the prior for α . Conditioning on α and $(x_1, r_1, \ldots, r_{N-1})$, we can sample from the truncated gamma density $f_{\alpha, x_1 r_1 \ldots r_{N-1}}(y)$ using the exact simulation algorithm in Appendix A. Thus, we choose the envelope

$$C'' \alpha^{N+\nu_1-1} \times \exp\left(-\alpha \left(\nu_2 + \int_{x_1 r_1 \dots r_{N-1}}^{\infty} w^{-1} e^{-w} dw\right)\right) \times f_{\alpha, x_1 r_1 \dots r_{N-1}}(y),$$

where C'' is a scaling constant. To sample from the envelope, we first simulate a gamma random number

$$\tilde{\alpha} \sim \operatorname{Ga}\left(N + \nu_1, \nu_2 + \int_{x_1 r_1 \dots r_{N-1}}^{\infty} w^{-1} e^{-w} dw\right),$$

then sample from $f_{\tilde{\alpha}, x_1 r_1...r_{N-1}}(\tilde{y})$. The ratio between the target density and the envelope is maximised at

$$\frac{C'}{C''}\frac{y^{n_{N+1}}}{(\Sigma+y)^n} \leq \frac{C'}{C''}\frac{n_{N+1}^{n_{N+1}}}{n^n}\frac{(n-n_{N+1})^{n-n_{N+1}}}{(\Sigma)^{n-n_{N+1}}},$$

thus, the candidates $\tilde{\alpha}$ and \tilde{y} are accepted if

$$U[0, 1] \leq \left(\frac{n}{n_{N+1}} \frac{\tilde{y}}{\Sigma + \tilde{y}}\right)^{n_{N+1}} \left(\frac{n}{n - n_{N+1}} \frac{\Sigma}{\Sigma + \tilde{y}}\right)^{n - n_{N+1}}$$

and this completes Step 2.

The quantity $\tilde{y}/(\Sigma + \tilde{y})$ on the right hand side of the inequality above represents the tail probability of a truncated Poisson–Dirichlet process, and n_{N+1} stands for the number of latent variables in this cluster. When the truncation level N is large enough, we expect both quantities to be small, and the acceptance rate should be high. If the truncation level N is not large enough, these quantities could be larger. But since the envelope sampling algorithm is very efficient, the acceptance–rejection method may still be feasible. In more extreme cases, we could include y and α in the HMC sampler. To this end, we include $Y := \log(y)$ and $A := \log(\alpha)$ in (13) to obtain unconstrained variables.

Finally, we output the accepted candidates of $(x_1, r_1, ..., r_{N-1}, y)$ and α . The posterior jump sizes of the gamma process are derived from $J_1 = x_1$, $J_2 = x_1r_1$, ..., $J_N = x_1r_1 \dots r_{N-1}$, and the posterior random weights are given by

$$p_1 = \frac{J_1}{\Sigma + y}, \quad \dots, \quad p_N = \frac{J_N}{\Sigma + y}, \quad r = \frac{y}{\Sigma + y},$$

where $\Sigma = J_1 + \cdots + J_N$.

From the output we can see that the truncated Poisson– Dirichlet process differs from the almost sure truncation (4) in the following two aspects. First, we include the largest Nrandom weights in the posterior inference, while the almost sure truncation is based on arbitrary N random weights. The largest random weights would be helpful if we want to focus on the most important components in the model. Second, our method is targetted at the jump sizes of the underlying gamma process, which are not available in the almost sure truncation. The jump sizes are critical to some researches, for example the Caron–Fox network model, and the truncated Poisson–Dirichlet process should be used in this case.

6 Models based on truncated Poisson–Dirichlet process

In this section, we consider two models based on the truncated Poisson–Dirichlet process, and discuss their posterior inference method.

Consider a Bayesian nonparametric hierarchical model of the format

$$(X_i \mid Y_i) \sim \pi(X_i \mid Y_i), \quad i = 1, \dots, n,$$

$$(Y_i \mid P) \sim P,$$

$$P \sim \mathcal{P}.$$
(15)

In this model, X_1, \ldots, X_n represent the observations, they are assumed to be independent conditionally on the latent variables Y_1, \ldots, Y_n . The latent variables Y_i are independent and identically distributed according to P, and P is a sample of the random probability measure \mathcal{P} . We refer to \mathcal{P} as the model prior.

If we use a Dirichlet process for \mathcal{P} , (15) reverts to a Dirichlet process hierarchical model. Since the Dirichlet process prior is conjugate to multinomial sampling, the model posterior can be obtained by integrating over P and exploiting the Pólya urn scheme. This approach leads to a marginalised version of the Bayesian nonparametric hierarchical model:

$$(X_i \mid Y_i) \sim \pi(X_i \mid Y_i), \quad i = 1, \dots, n,$$

$$(Y_1, \dots, Y_n) \sim \pi_{\infty}(Y_1, \dots, Y_n),$$

$$\pi_{\infty}(dY_1, \dots, dY_n) \sim H(dY_1) \prod_{i=2}^n \left\{ \frac{\alpha}{\alpha + i - 1} H(dY_i) + \frac{1}{\alpha + i - 1} \sum_{j=1}^{i-1} \delta_{Y_j}(dY_i) \right\},$$

where $\pi_{\infty}(dY_1, \ldots, dY_n)$ denotes the distribution of a Pólya urn scheme.

The marginalisation method is widely used in Bayesian nonparametric statistics as it can be implemented easily. However, since the Dirichlet process prior is integrated out by marginalisation, this approach has the side effect of making inference based on the latent variables Y_1, \ldots, Y_n only. To combat this problem, Ishwaran and Zarepour (2000) (see also Ishwaran and James 2002) replaced \mathcal{P} by the almost sure truncation $\tilde{\mathcal{P}}_N$ (see Eq. 4). This leads to an approximation for the Dirichlet process hierarchical model which is almost indistinguishable from its limit when N is large enough. Moreover, a direct inference for $\tilde{\mathcal{P}}_N$ can be obtained using this method.

Following this idea, we provide a new approximation for model (15). Our method is to replace the Dirichlet process prior \mathcal{P} by a truncated Poisson–Dirichlet process \mathcal{P}_N . To facilitate the posterior sampling algorithm, we recast the

model completely in terms of random variables, then model (15) is approximated by

$$(X_{i} | \mathbf{Z}, \mathbf{K}) \sim \pi(X_{i} | Z_{K_{i}}), \quad i = 1, \dots, n,$$

$$(K_{i} | P) \sim \sum_{i=1}^{N} p_{i}\delta_{i}(.) + r\delta_{N+1}(.),$$

$$P \sim \mathcal{P}_{N},$$

$$\mathbf{Z} \sim \pi(\mathbf{Z}),$$

(16)

where $\mathbf{K} = (K_1, ..., K_n)$ are the classifiers that relate the random variables $\mathbf{Z} = (Z_1, ..., Z_n)$ to the latent variables Y_i , i.e., $Y_i = Z_{K_i}$. We will devise a blocked Gibbs sampler to explore the posterior distribution of model (16), denoted by $\pi(P, \mathbf{Z}, \mathbf{K} | \mathbf{X})$. The sampler draws iteratively from the conditional distributions $\pi(P, \mathbf{Z} | \mathbf{K}, \mathbf{X})$ and $\pi(\mathbf{K} | P, \mathbf{Z}, \mathbf{X})$, thus producing the posterior values $(P^{(b)}, \mathbf{Z}^{(b)}, \mathbf{K}^{(b)})$.

Next, we use the normal mean mixture model to illustrate the posterior sampling algorithm. These models are in the format of (15) with $(X_i | Y_i) \sim \mathcal{N}(Y_i, \sigma_X)$, where σ_X denotes the variance. In the clustering problems, σ_X is assumed to be a known constant; while in the predictive density estimation problems, σ_X is a parameter that needs to be estimated. For now we assume that σ_X is known; this condition will be released to obtain a Bayesian semiparametric hierarchical model later. Using the approximation (16), we express the normal mean mixture model as

$$(X_{i} | \mathbf{Z}, \mathbf{K}) \sim \mathcal{N}(Z_{K_{i}}, \sigma_{X}), \quad i = 1, \dots, n,$$

$$(K_{i} | P) \sim \sum_{i=1}^{N} p_{i}\delta_{i}(.) + r\delta_{N+1}(.),$$

$$(Z_{k} | \theta, \sigma_{Z}) \sim \mathcal{N}(\theta, \sigma_{Z}), \quad (17)$$

$$(\theta | \sigma_{\theta}) \sim \mathcal{N}(0, \sigma_{\theta}),$$

$$(\sigma_{Z}^{-1} | \tau_{1}, \tau_{2}) \sim \operatorname{Ga}(\tau_{1}, \tau_{2}),$$

$$(\alpha | \nu_{1}, \nu_{2}) \sim \operatorname{Ga}(\nu_{1}, \nu_{2}),$$

where *P* is a sample of the truncated Poisson–Dirichlet process \mathcal{P}_N .

In model (17), we use hyperparameters θ and σ_Z for the distribution of Z_k , and choose normal and inverse gamma priors for these hyperparameters respectively. Since the priors are conjugate, the hyperparameters can be updated easily. To obtain noninformative priors, we choose a large value for σ_{θ} , i.e., $\sigma_{\theta} = 1000$, and use small values for τ_1 and τ_2 , i.e., $\tau_1 = \tau_2 = 0.001$. Note that α is directly related to the number of distinct clusters. A large α encourages the mass to spread among distinct K_i values, while a small α means the mass will concentrate on a few clusters. We choose a gamma prior for α due to its flexibility, as we can manipulate the number of distinct clusters by adjusting the hyperparameters ν_1

and ν_2 . More importantly, the gamma prior will simplify the update procedure for α in the posterior sampling algorithm.

Next, we design a Gibbs sampling scheme for the posterior of (17). To implement the blocked Gibbs sampler, we need to draw from the following conditional distributions iteratively:

$$(P, \alpha \mid \mathbf{K}), \tag{18}$$

 $(\mathbf{Z} \mid \mathbf{K}, \theta, \sigma_Z, \mathbf{X}), \tag{19}$

 $(\mathbf{K} \mid P, \mathbf{Z}, \mathbf{X}), \tag{20}$

$$(\theta \mid \mathbf{Z}, \sigma_Z), \tag{21}$$

$$(\sigma_Z \mid \mathbf{Z}, \theta). \tag{22}$$

The first step (18) samples from the posterior of the truncated Poisson–Dirichlet process, then updates the concentration parameter. This can be done using the algorithm introduced in Sect. 5. To sample from (19), (20), (21) and (22), we follow the standard procedure in literature. The means \mathbf{Z} can be updated using the conditional density

$$f(\mathbf{Z} \mid \mathbf{K}, \theta, \sigma_{Z}, \mathbf{X})$$

$$\propto \left\{ \prod_{j=1}^{m} f(Z_{K_{j}^{*}} \mid \theta, \sigma_{Z}) \prod_{\{i: K_{i} = K_{j}^{*}\}} f(X_{i} \mid Z_{K_{j}^{*}}) \right\}$$

$$\times f(Z^{K} \mid \theta, \sigma_{Z}),$$

where $\{K_1^*, \ldots, K_m^*\}$ denote the *m* unique values in the set **K**. For each element of $\{K_1^*, \ldots, K_m^*\}$, we sample from $Z_{K_j^*}$ according to

$$(Z_{K_j^*} | \mathbf{K}, \theta, \sigma_Z, \mathbf{X}) \sim \mathcal{N}(\theta_j^*, \sigma_{Z_j}^*),$$

where $\theta_j^* = \sigma_{Z_j}^* (\theta/\sigma_Z + \sum_{\{i:K_i=K_j^*\}} X_i/\sigma_X), \ \sigma_{Z_j}^* = (n_j/\sigma_X + 1/\sigma_Z)^{-1}$, and n_j is the number of times K_j^* occurs in **K**. Also, for each $j \in \mathbf{K} - \{K_1^*, \ldots, K_m^*\}$, we sample from $Z_j \sim \mathcal{N}(\theta, \sigma_Z)$ independently.

The conditional distribution for K is given by

$$(K_j \mid P, \mathbf{Z}, \mathbf{X}) \sim \sum_{i=1}^N p_{i,j}^* \delta_i(.) + r_j^* \delta_{N+1}(.),$$

where

$$(p_{1,j}^*, \dots, p_{N,j}^*, r_j^*) \\ \propto \left(p_1 \exp\left(\frac{-1}{2\sigma_X} (X_j - Z_1)^2\right), \dots, p_N \exp\left(\frac{-1}{2\sigma_X} (X_j - Z_N)^2\right), \dots, r \exp\left(\frac{-1}{2\sigma_X} (X_j - Z_{N+1})^2\right) \right).$$

The conditional distribution of θ and σ_Z are conjugate. We sample from θ according to $(\theta \mid \mathbf{Z}, \sigma_Z) \sim \mathcal{N}(\theta^*, \sigma_{\theta}^*)$, where

$$\theta^* = \frac{\sigma_{\theta}^*}{\sigma_Z} \sum_{i=1}^{N+1} Z_i \text{ and } \sigma_{\theta}^* = (N/\sigma_Z + 1/\sigma_{\theta})^{-1}.$$

And for the conditional distribution of σ_Z , we sample from

$$(\sigma_Z^{-1} | \mathbf{Z}, \theta) \sim \text{Ga}\left(\tau_1 + N/2, \tau_2 + \sum_{i=1}^{N+1} (Z_i - \theta)^2/2\right).$$

These steps are taken from Ishwaran and James (2002) and Ishwaran and Zarepour (2000), they complete all the steps of the blocked Gibbs sampler and enables us to sample from the posterior of model (17).

The blocked Gibbs sampler can be extended to estimate the variance of the underlying normal distribution. By viewing the variance σ_X as an unknown parameter, we get a Bayesian semiparametric hierarchical model where $(X_i | Y_i, \sigma_X) \sim \mathcal{N}(Y_i, \sigma_X)$. We choose an inverse gamma prior for σ_X :

$$\sigma_X^{-1} \sim \operatorname{Ga}(\gamma_1, \gamma_2) \tag{23}$$

with $\gamma_1 = \gamma_2 = 0.001$, such that the prior is noninformative. Then σ_X can be updated by conjugation

$$(\sigma_X^{-1} \mid \mathbf{X}, \mathbf{Z}, \mathbf{K})$$

~ $\operatorname{Ga}\left(\gamma_1 + n/2, \gamma_2 + \sum_{i=1}^n (X_i - Z_{K_i})^2/2\right).$ (24)

We add an extra step (24) into the blocked Gibbs sampler, and use the updated value of σ_X in other steps. This gives us a sampler for the normal mean mixture model with an unknown variance.

The normal mean mixture model demonstrates the feasibility of the truncated Poisson–Dirichlet process in clustering and predictive density estimation problems. However, these problems can also be solved using the almost sure truncation, and it still remains to justify the novelty of our method. To this end, we provide an application of the truncated Poisson– Dirichlet process in the Caron–Fox network model.

As discussed in Sect. 3, the construction of the truncated Poisson–Dirichlet process is partially motivated by the 'network with key nodes', meaning that we want to estimate the largest N popularity parameters in the network. The twodimensional Pólya urn scheme in Williamson (2016) is not applicable in this problem because the popularity parameters are integrated out by marginalisation; while the method in Caron and Fox (2017) hinges on the estimation of the popularity parameters of all the nodes with at least one link. To our best knowledge, no literature has focused on the most popular nodes in a Caron–Fox network model. We replace the Dirichlet process prior in model (7) by a truncated Poisson–Dirichlet process \mathcal{P}_N and consider the network model

$$G \sim \mathcal{P}_{N}$$

$$s_{j}, r_{j} \stackrel{\text{i.i.d.}}{\sim} G = \sum_{i=1}^{N} \frac{J_{i}}{\tau_{\alpha}} \delta_{i}(.) + \frac{\tau_{\alpha}^{N-}}{\tau_{\alpha}} \delta_{N+1}(.), \qquad (25)$$

for j = 1, ..., n, where $\{J_i\}_{i=1}^{\infty}$ are the ranked jumps of a gamma process τ_{α} , they denote the popularity parameters of the nodes in the network; and $\tau_{\alpha}^{N-} = \sum_{i=N+1}^{\infty} J_i$ is the total popularity parameter of the less popular nodes.

Next, we discuss the posterior inference procedure for model (25). For a set of observations from the network, we need to count the number of links of all the nodes and find out the largest *N* values, namely n_1, \ldots, n_N ; they are explained as the number of links of the *N* nodes with the highest popularity parameters. The other links are grouped together in terms of $n_{N+1} := \sum_{i=N+1}^{\infty} n_i$ and understood as the total number of links of the less popular nodes. Then we input the data $\{n_1, \ldots, n_N, n_{N+1}\}$ into the posterior of model (25), which is given in Proposition 4, and use the algorithm in Sect. 5 to sample from it. Note that we will only output the jump sizes $J_1, \ldots, J_n, \tau_{\alpha}^{N-}$ in this case.

We emphasise that this method estimates the actual popularity parameters $(J_1, \ldots, J_N, \tau_{\alpha}^{N-})$, not only their weights, and this makes the model more flexible. For example, to include the Poisson assumption about the total number of links, we only need to multiply the posterior distribution (12) by a Poisson probability mass function. See Caron and Fox (2017) for a further discussion on the choice of the prior for α , as well as the applications of the posterior inference results.

7 Numerical implementations

In this section, we present some numerical results for the models introduced in Sect. 6. We first study the normal mean mixture model with a known variance. Consider a mixture of three normal distributions whose means are $\{-3, 1, 2\}$ with equal probabilities and variance $\sigma_X = 1$, we simulate n = 45 observations from the model, then run the blocked Gibbs sampler and record the posterior mean values. We use the truncation level N = 45 for the truncated Poisson–Dirichlet process prior, and choose a Ga(2, 2) prior for α . In the HMC step, we use the leapfrog steps L = 10 and the step size $\epsilon = 0.25$. The numerical results are presented in Fig. 3.

The outcome of the blocked Gibbs sampler can be used to estimate the predictive density for a new observation X_{n+1} . We denote by $f(X_{n+1} | \mathbf{X})$ the predictive density for X_{n+1} conditioning on the current observations $\mathbf{X} = (X_1, \dots, X_n)$, 30



Fig. 3 Posterior mean values based on 45 observations, each observation X_i is sampled from a normal mean mixture model with mean $\{-3, 1, 2\}$ with equal probabilities and variance $\sigma_X = 1$. Values for X_i and true values for μ_X are denoted by green cross and red plus respectively. The plot is based on 4500 sampled values following an initial 2500 iteration burn-in using the blocked Gibbs sampler

and Y_{n+1} the latent variable of this new observation, then

$$f(X_{n+1} | \mathbf{X})$$

$$= \int \mathcal{N}(X_{n+1} | Y_{n+1}, \sigma_X) d\pi(Y_{n+1}, \sigma_X | \mathbf{X})$$

$$= \iint \mathcal{N}(X_{n+1} | Y_{n+1}, \sigma_X)$$

$$\times d\pi(Y_{n+1} | P) d\pi(P, \sigma_X | \mathbf{X}).$$

Since we have used a truncated Poisson–Dirichlet process prior, the probability measure *P* has the format $P(.) = \sum_{i=1}^{N} p_i \delta_{Z_i}(.) + r \delta_{Z_{N+1}}(.)$, and the inner integral can be expressed as

$$\int \mathcal{N}(X_{n+1} | Y_{n+1}, \sigma_X) d\pi(Y_{n+1} | P)$$

$$= \sum_{i=1}^{N} p_i \mathcal{N}(X_{n+1} | Z_i, \sigma_X)$$

$$+ r \mathcal{N}(X_{n+1} | Z_{N+1}, \sigma_X).$$
(26)

Thus, the predictive density $f(X_{n+1} | \mathbf{X})$ can be estimated by averaging (26) over the posterior values from different iterations. In Fig. 4, we randomly select fifty iterations and use their posterior values to evaluate (26). From the figure we can see that the posterior induces two modes for the model, but it cannot distinguish between the two modes 1 and 2 which are close to each other.

The performance of the blocked Gibbs sampler can also be seen from the clustering of observations, which is revealed



Fig. 4 Fifty randomly selected predictive densities of the format (26). The true values for X_i are presented by the histogram. The plot is based on 4500 sampled values following an initial 2500 iteration burn-in using the blocked Gibbs sampler



Fig. 5 Proportions of the number of distinct values in $\mathbf{K}^{(b)}$, $b = 1, \dots, 4500$. N = 45

by the number of distinct values in $\mathbf{K}^{(b)}$. We count the number of distinct values from each iteration and record their proportions in Fig. 5. The figure shows that the posterior distribution has at least two clusters, and it is unlikely to have more than ten clusters. Then we re-run the blocked Gibbs sampler with a higher truncation level N = 250 for the truncated Poisson–Dirichlet process prior, and count the number of distinct values in $\mathbf{K}^{(b)}$ again. From Fig. 6 we can see that the proportions are similar as before, although the posterior distribution has at most eight clusters in this case.

For real-world data sets, we also need to estimate the variance σ_X , and this leads us to the normal mean mixture model with an unknown variance. Next, we provide an illustration based on the galaxy velocities data in Roeder (1990). The data contains the relative velocities of 82 galaxies from six well-



Fig. 6 Proportions of the number of distinct values in $\mathbf{K}^{(b)}$, $b = 1, \dots, 4500$. N = 250



Fig. 7 Posterior mean values (km/sec. * 0.001) of the galaxy velocities data. The observations X_i are denoted by green cross. The plot is based on 2500 sampled values following an initial 2500 iteration burn-in using the blocked Gibbs sampler with N = 82 and $\alpha \sim Ga(2, 4)$

separated conic sections of space. Escobar and West (1995) analysed the data using a Dirichlet process mixture model, while Ishwaran and Zarepour (2000) replaced the Dirichlet process by the almost sure truncation (see Eq. 4). We reanalyse the data using the normal mean mixture model (17) with an unknown variance, and record the posterior mean values in Fig. 7. Then we estimate the predictive density (26) using the outcome of the blocked Gibbs sampler and present the numerical results in Fig. 8. We can see that there seems to be six distinct modes in the predictive density. This finding is in line with the results of Escobar and West (1995) and Ishwaran and Zarepour (2000).

We provide another illustration for the normal mean mixture model with unknown variance using the Mexican stamp



Fig.8 Twenty-five randomly selected predictive densities of the format (26). The densities are evaluated over the same partition



Fig. 9 Posterior mean values (mm * 100) of the stamp thickness data. The observations X_i are denoted by green cross. The plot is based on 2500 sampled values following an initial 2500 iteration burn-in using the blocked Gibbs sampler with N = 150 and $\alpha \sim Ga(2, 2)$

thickness data. The 1872–1874 Hidalgo postage stamps of Mexico were printed on handmade paper, and their thickness varies considerably. Izenman and Sommer (1988) tested the number of modes in thickness by analyzing 485 thickness data and concluded that there were seven modes, namely 0.072, 0.080, 0.090, 0.100, 0.110, 0.120 and 0.130 mm. Ishwaran and James (2002) used a normal mean mixture model with the almost sure truncation of Dirichlet process to estimate the predictive density. We revisit the data using model (17) with an unknown variance, and record the numerical results in Figs. 9 and 10. From the figures we can recognise seven distinct modes in the predictive density, and the values of the modes are similar to the results of Izenman and Sommer (1988).



Fig. 10 Fifty randomly selected predictive densities of the format (26). The densities are evaluated over the same partition

| Index of receiving node | 1 | 541 | 329 | 204 | 161 | 111 | 75 | 69 | 66 | 43 | 29 | 185 |
|-------------------------|-----|-----|-----|-----|-----|-----|----|----|----|----|----|------|
| | 2 | 282 | 162 | 124 | 85 | 71 | 52 | 46 | 30 | 28 | 15 | 92 |
| | 3 | 190 | 109 | 85 | 61 | 57 | 32 | 18 | 20 | 18 | 18 | 81 |
| | 4 | 156 | 92 | 51 | 44 | 29 | 28 | 14 | 17 | 11 | 13 | 53 |
| | 5 | 127 | 55 | 46 | 39 | 25 | 23 | 16 | 11 | 7 | 9 | 28 |
| | 6 | 84 | 54 | 41 | 34 | 23 | 12 | 11 | 7 | 4 | 8 | 19 |
| | 7 | 72 | 40 | 30 | 19 | 15 | 7 | 13 | 9 | 5 | 2 | 26 |
| | 8 | 62 | 24 | 31 | 21 | 13 | 8 | 8 | 7 | 3 | 2 | 16 |
| | 9 | 56 | 26 | 18 | 10 | 2 | 9 | 8 | 5 | 4 | 0 | 15 |
| | 10 | 33 | 18 | 17 | 17 | 10 | 8 | 5 | 4 | 5 | 0 | 15 |
| r | est | 178 | 95 | 72 | 60 | 32 | 31 | 20 | 21 | 17 | 10 | 66 |
| | | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 | rest |
| Index of sending node | | | | | | | | | | | | |

Fig. 11 Simulated observation of 6000 links, each link is sampled from the truncated Poisson–Dirichlet process network model (25) with truncation level N = 10 and concentration parameter $\alpha = 5$

Next, we provide an numerical implementation for the Caron–Fox network model. We first generate simulated observations from model (25). Using the algorithm introduced in Sect. 4, we draw a sample *G* from the model prior \mathcal{P}_N . Then we choose the origins and destinations for the links according to *G*. Since each link involves two nodes, we need to draw 2n independent samples from *G*. We record the simulated observations in Fig. 11. Note that the number in each box denotes the amount of links with the same origin and destination. Then we apply the posterior sampling algorithm to estimate the popularity parameters of the nodes. The numerical results are presented in Fig. 12. The figure shows that our method recovers the popularity parameters of the prior very well.



Fig. 12 Posterior values of the popularity parameters. The true values of $(J_1, \ldots, J_N, \tau_{\alpha}^{N-})$ are denoted by green cross. The plot is based on 2500 sampled values following an initial 2500 iteration burn-in using the blocked Gibbs sampler with N = 10 and $\alpha \sim \text{Ga}(0.001, 0.001)$

Finally, we investigate the complexity of our posterior sampling algorithm, and compare it to complexity of the blocked Gibbs sampler based on almost sure truncation, which can be found in Ishwaran and Zarepour (2000). Here we explain the complexity as the total number of random variables generated by the algorithm, which contributes to the majority of the computation time. For the normal mean mixture model, our posterior sampler differs from the almost sure truncation method only in step (18), hence we only record the complexity of this step. While for the network model, the almost sure truncation is not applicable. In Table 2, we present the complexities of different algorithms. From the table we can see that our sampler generates more random variables than the almost sure truncation method. In fact, the truncated Poisson-Dirichlet process is not expected to outperform the existing methods based on the almost sure truncation. The purpose of our method is to include the actual jump sizes $(J_1, \ldots, J_N, \tau_{\alpha}^{N-})$ into estimation, and that requires more computation.

8 Discussion

In this research, we construct a new random probability measure called the truncated Poisson–Dirichlet process, and use it to approximate the distribution of the Dirichlet process. Then we develop a posterior inference method for the truncated Poisson–Dirichlet process, and illustrate its usage in the normal mean mixture model and Caron–Fox network model.

There are several topics worth a further study. First, our truncation method can be easily extended to the posterior of a Dirichlet process. This means the standard Dirichlet process can be used as the prior, and the posterior will be truncated instead. A similar idea in literature was mentioned in Sect. 3. Another topic is to generalise our truncation method to the Pitman-Yor process and generalised gamma process, this will make use of the algorithms introduced by Dassios and Zhang (2021) and Dassios et al. (2020). Moreover, we will investigate the application of our method in the analysis of real-world network data sets, where the generalised gamma process will be considered. Finally, the truncated Poisson-Dirichlet process seems appropriate for the Bayesian nonparametric Plackett-Luce model (see Caron and Teh 2012; Caron et al. 2014), and we will study it in a future work.

Acknowledgements The authors are grateful to the editor and referee for giving us extremely helpful and inspiring comments.

Funding The authors have no relevant financial or non-financial interests to disclose.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecomm ons.org/licenses/by/4.0/.

Table 2Complexity of theposterior sampling algorithmbased on different priors, ASTand TPD stand for the almostsure truncation and truncatedPoisson–Dirichlet processrespectively, CPU is at 2.80 GHz

| Problem | Prior | Ν | Iterations | Uniform | Beta | Gamma | Normal | Time (s) |
|-------------|-------|------|------------|---------|---------|-------|--------|----------|
| Normal mean | AST | 45 | 7000 | 0 | 308000 | 7000 | 0 | 4.3541 |
| Estimation | TPD | 45 | 7000 | 93436 | 21812 | 7000 | 315000 | 96.9472 |
| Galaxy | AST | 82 | 5000 | 0 | 405000 | 5000 | 0 | 5.5732 |
| Velocity | TPD | 82 | 5000 | 70871 | 16957 | 5000 | 410000 | 147.1075 |
| Stamp | AST | 150 | 5000 | 0 | 745000 | 5000 | 0 | 29.9868 |
| Thickness | TPD | 150 | 5000 | 76196 | 18732 | 5000 | 750000 | 487.1008 |
| Network | AST | N.A. | N.A. | N.A. | N.A. | N.A. | N.A. | N.A. |
| Analysis | TPD | 10 | 5000 | 5724797 | 1782282 | 85891 | 50000 | 96.7071 |
| | | | | | | | | |

Appendix A: Simulation of truncated gamma process

In this appendix, we attach Algorithm 3.2 of Dassios et al. (2019). This algorithm exactly samples from a truncated gamma process $Z_{t,\mu}$ with Lévy–Khintchine representation

$$\mathbb{E}(\exp(-\beta Z_{t,\mu}))$$

= $\exp\left(-t\int_0^1(1-e^{-\beta w})w^{-1}e^{-\mu w}dw\right).$

Algorithm 2 Exact simulation of $Z_{t,\mu}$

1: Set S = 0.

2: Numerically minimise

 $C(v, \delta)$

$$= \frac{\Gamma(\delta)e^{-\mu}}{\nu(1-\delta)e}e^{\Gamma(0,\mu)\exp(\Gamma(0,\mu)+\nu+\log(\mu))}$$
$$\times \mu^{\exp(\Gamma(0,\mu)+\nu+\log(\mu))}$$

- $\times e^{\nu \exp(\Gamma(0,\mu)+\nu+\log(\mu))}$
- × $(\Gamma(\exp(\Gamma(0,\mu) + \nu + \log(\mu)) + \delta))^{-1}$,

record the optimal value ν^* and δ^* , and set $C = C(\nu^*, \delta^*)$. 3: Generate $U_1 \sim U[0, 1]$ and set $T = -\log(U_1)/\nu^*$.

4: Generate $Y \sim \text{Beta}(T, \delta^*), U_2 \sim U[0, 1], \text{ and set}$

$$M = Y - 1 + (1 - Y) \exp(-U_2 \log(1 - Y)).$$

5: Generate $V_1 \sim U[0, 1]$, if

$$V_{1} \leq \frac{1}{C} \frac{\Gamma(\delta^{*})}{\nu^{*}} \frac{e^{\Gamma(0,\mu)T} \mu^{T} e^{\nu^{*}T}}{\Gamma(T+\delta^{*})} \\ \times \frac{-\log(1-Y)}{(1-Y)^{\delta^{*}-1}} e^{-\mu(1+M)},$$

then accept (T, M). Otherwise, go to Step 3.

- 6: If T > t, continue to Step 7. Otherwise, set S = S+1+M, t = t-T, and go to Step 3.
- 7: Generate $U_3 \sim U[0, 1]$ and set $X = U_3^{1/t}$. 8: Generate $V_2 \sim U[0, 1]$. If

 $V_2 \le \exp(-\mu X),$

then accept X. Otherwise, go to Step 7. 9: Return S + X.

References

Al Labadi, L., Zarepour, M.: On simulations from the two-parameter Poisson-Dirichlet process and the normalized inverse-Gaussian process. Sankhya A 76(1), 158–176 (2014) Argiento, R., Guglielmi, A., Pievatolo, A.: Bayesian density estimation and model selection using nonparametric hierarchical mixtures. Comput. Stat. Data Anal. 54(4), 816–832 (2010)

Page 19 of 20

30

- Argiento, R., Bianchini, I., Guglielmi, A.: A blocked Gibbs sampler for NGG-mixture models via a priori truncation. Stat. Comput. 26(3), 641–661 (2016)
- Barrios, E., Lijoi, A., Nieto-Barajas, L.E., et al.: Modeling with normalized random measure mixture models. Stat. Sci. 28(3), 313–334 (2013)
- Blackwell, D., MacQueen, J.B.: Ferguson distributions via Pólya urn schemes. Ann. Stat. 1, 353–355 (1973)
- Campbell, T., Huggins, J.H., How, J.P., et al.: Truncated random measures. Bernoulli 25(2), 1256–1288 (2019)
- Caron, F., Fox, E.B.: Sparse graphs using exchangeable random measures. J. R. Stat. Soc. Ser. B Stat. Methodol. 79(5), 1295–1366 (2017)
- Caron, F., Teh, Y.: Bayesian nonparametric models for ranked data. In: Advances in Neural Information Processing Systems, vol. 25 (2012)
- Caron, F., Teh, Y.W., Murphy, T.B.: Bayesian nonparametric Plackett– Luce models for the analysis of preferences for college degree programmes. Ann. Appl. Stat. 8(2), 1145–1181 (2014)
- Chi, Z.: On exact sampling of nonnegative infinitely divisible random variables. Adv. Appl. Probab. 44(3), 842–873 (2012)
- Cloud, K., Huber, M.: Fast perfect simulation of Vervaat perpetuities. J. Complex. 42, 19–30 (2017)
- Connor, R.J., Mosimann, J.E.: Concepts of independence for proportions with a generalization of the Dirichlet distribution. J. Am. Stat. Assoc. 64, 194–206 (1969)
- Dassios, A., Zhang, J.: Exact simulation of two-parameter Poisson-Dirichlet random variables. Electron. J. Probab. 26, 1–20 (2021)
- Dassios, A., Qu, Y., Lim, J.W.: Exact simulation of generalised Vervaat perpetuities. J. Appl. Probab. 56(1), 57–75 (2019)
- Dassios, A., Lim, J.W., Qu, Y.: Exact simulation of a truncated Lévy subordinator. ACM Trans. Model. Comput. Simul. 30(3), 1–17 (2020)
- Devroye, L., Fawzi, O.: Simulating the Dickman distribution. Stat. Probab. Lett. **80**(3–4), 242–247 (2010)
- Escobar, M.D.: Estimating normal means with a Dirichlet process prior. J. Am. Stat. Assoc. **89**(425), 268–277 (1994)
- Escobar, M.D., West, M.: Bayesian density estimation and inference using mixtures. J. Am. Stat. Assoc. 90(430), 577–588 (1995)
- Ferguson, T.S.: A Bayesian analysis of some nonparametric problems. Ann. Stat. 1, 209–230 (1973)
- Ferguson, T.S., Klass, M.J.: A representation of independent increment processes without Gaussian components. Ann. Math. Stat. 43, 1634–1643 (1972)
- Fill, J.A., Huber, M.L.: Perfect simulation of Vervaat perpetuities. Electron. J. Probab. 15(4), 96–109 (2010)
- Gelfand, A.E., Kottas, A.: A computational approach for full nonparametric Bayesian inference under Dirichlet process mixture models. J. Comput. Graph. Stat. 11(2), 289–305 (2002)
- Godsill, S., Kı ndap, Y.: Point process simulation of generalised inverse Gaussian processes and estimation of the Jaeger integral. Stat Comput 32(1), 1–18 (2022)
- Griffin, J.E.: An adaptive truncation method for inference in Bayesian nonparametric models. Stat. Comput. 26(1–2), 423–441 (2016)
- Horváth, G., Horváth, I., Almousa, S.A.D., et al.: Numerical inverse Laplace transformation using concentrated matrix exponential distributions. Perform. Eval. 137(102), 067 (2020)
- Imai, J., Kawai, R.: Numerical inverse Lévy measure method for infinite shot noise series representation. J. Comput. Appl. Math. 253, 264– 283 (2013)
- Ishwaran, H., James, L.F.: Gibbs sampling methods for stick-breaking priors. J. Am. Stat. Assoc. 96(453), 161–173 (2001)

- Ishwaran, H., James, L.F.: Approximate Dirichlet process computing in finite normal mixtures: smoothing and prior information. J. Comput. Graph. Stat. **11**(3), 508–532 (2002)
- Ishwaran, H., Zarepour, M.: Markov chain Monte Carlo in approximate Dirichlet and beta two-parameter process hierarchical models. Biometrika **87**(2), 371–390 (2000)
- Izenman, A.J., Sommer, C.J.: Philatelic mixtures and multimodal densities. J. Am. Stat. Assoc. 83(404), 941–953 (1988)
- James, L.F., Lijoi, A., Prünster, I.: Posterior analysis for normalized random measures with independent increments. Scand. J. Stat. 36(1), 76–97 (2009)
- Kingman, J.F.C.: Random discrete distributions. J. R. Stat. Soc. Ser. B Stat. Methodol. 37(1), 1–15 (1975)
- Kyprianou, A.E.: Introductory Lectures on Fluctuations of Lévy Processes with Applications. Universitext, Springer, Berlin (2006)
- McCloskey, J.W.: A Model for the Distribution of Individuals by Species in an Environment. Michigan State University, Department of Statistics (1965)
- Muliere, P., Tardella, L.: Approximating distributions of random functionals of Ferguson–Dirichlet priors. Canad. J. Stat. 26(2), 283–297 (1998)
- Neal, R.M.: MCMC using Hamiltonian dynamics. In: Handbook of Markov chain Monte Carlo. Chapman & Hall/CRC Handb. Mod. Stat. Methods, CRC Press, Boca Raton, FL, pp. 113–162 (2011)
- Perman, M., Pitman, J., Yor, M.: Size-biased sampling of Poisson point processes and excursions. Probab. Theory Related Fields 92(1), 21–39 (1992)
- Pitman, J.: Random discrete distributions invariant under size-biased permutation. Adv. Appl. Probab. 28(2), 525–539 (1996)
- Pitman, J.: Some developments of the Blackwell-MacQueen urn scheme. In: Statistics, Probability and Game Theory, IMS Lecture Notes Monogr. Ser., vol 30. Inst. Math. Statist., Hayward, CA, pp 245–267 (1996b)
- Pitman, J.: Combinatorial Stochastic Processes, Lecture Notes in Mathematics, vol. 1875. Springer, Berlin (2006)

- Roeder, K.: Density estimation with confidence sets exemplified by superclusters and voids in the galaxies. J. Am. Stat. Assoc. **85**(411), 617–624 (1990)
- Rosiński, J.: Series representations of Lévy processes from the perspective of point processes. In: Lévy processes. Birkhäuser Boston, Boston, MA, pp. 401–415 (2001)
- Sethuraman, J.: A constructive definition of Dirichlet priors. Stat. Sin. 4(2), 639–650 (1994)
- Williamson, S.A.: Nonparametric network models for link prediction. J. Mach. Learn. Res. 17, 7102–7121 (2016)
- Wolpert, R.L., Ickstadt, K.: Simulation of Lévy random fields. In: Practical Nonparametric and Semiparametric Bayesian Statistics, Lect. Notes Stat., vol 133. Springer, New York, pp. 227–242 (1998)
- Yamato, H.: Statistics Based on Dirichlet Processes and Related Topics. SpringerBriefs in Statistics, Springer, Singapore, JSS Research Series in Statistics (2020)
- Zarepour, M., Al Labadi, L.: On a rapid simulation of the Dirichlet process. Stat. Probab. Lett. **82**(5), 916–924 (2012)

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.